Author's response to reviews

Title: Information Transfer: What do decision-makers want and need from researchers

Authors:

- Maureen Dobbins (dobbinsm@mcmaster.ca)
- Peter Rosenbaum (rosenbau@mcmaster.ca)
- Nancy Plews (plewsn@mcmaster.ca)
- Mary Law (lawm@mcmaster.ca)
- Adam Fysh (fysha@mcmaster.ca)

Version: 4 Date: 22 June 2007

Author's response to reviews:

Dear editors, attached is a resubmission of the manuscript What decision makers want. I have made all the suggested changes to the text and tables as outlined in Martin's emails.